Author’s response to reviews

Title: Validation of the Multi-dimensional Scale of Perceived Social Support (MSPSS) and the relationship between social support, intimate partner violence and antenatal depression in Malawi.

Authors:

Robert C Stewart (robcstewart@mac.com)
Eric Umar (ericumar@yahoo.com)
Barbara Tomenson (barbara.tomenson@manchester.ac.uk)
Francis Creed (francis@fcreed.plus.com)

Version: 2 Date: 9 May 2014

Author’s response to reviews:

Dear Professor Roe,

Validation of the Multi-dimensional Scale of Perceived Social Support (MSPSS) and the relationship between social support, intimate partner violence and antenatal depression in Malawi.

Many thanks for the very helpful reviewers comments. Please find attached our detailed responses and the revised manuscript.

Many thanks for your help. I look forward to hearing from you.

Yours sincerely,

Dr Robert Stewart